

G OPEN ACCESS

Citation: Rayment Gomez S, Maddren R, Liyew EF, Chernet M, Anjulo U, Tamiru A, et al. (2024) Spatial heterogeneity in mass drug administration from a longitudinal epidemiological study assessing transmission interruption of soil transmitted helminths in the Wolaita zone of southern Ethiopia (Geshiyaro Project). PLoS Negl Trop Dis 18(2): e0011947. https://doi.org/10.1371/journal. pntd.0011947

Editor: David Joseph Diemert, George Washington University School of Medicine and Health Sciences, UNITED STATES

Received: July 4, 2023

Accepted: January 28, 2024

Published: February 8, 2024

Copyright: © 2024 Rayment Gomez et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying this article will be anonymised and shared on reasonable request to the Ethiopian Public Health Institute. Ethical permission from the Ethiopian government prevents data from being made freely available. Requests can be made to [https://ephi. gov.et/about-us/contact/]. RESEARCH ARTICLE

Spatial heterogeneity in mass drug administration from a longitudinal epidemiological study assessing transmission interruption of soil transmitted helminths in the Wolaita zone of southern Ethiopia (Geshiyaro Project)

Santiago Rayment Gomez^{1*}, Rosie Maddren¹, Ewnetu Firdawek Liyew², Melkie Chernet², Ufaysa Anjulo³, Adugna Tamiru³, Getachew Tollera², Geremew Tasew², Birhan Mengistu¹, Benjamin Collyer¹, Kathryn Forbes¹, Roy Anderson¹

1 London Centre for Neglected Tropical Disease Research, Department of Infectious Disease Epidemiology, Faculty of Medicine, St Marys Campus, Imperial College London, London, United Kingdom, 2 Bacterial, Parasitic and Zoonotic Disease Research Directorate, Ethiopian Public Health Institute, Addis Ababa, Ethiopia, 3 Disease Prevention and Health Promotion Core Process, Ministry of Health, Wolaita, Ethiopia

* santiago.rayment-gomez1@lshtm.ac.uk

Abstract

Objectives

Deworming programmes of soil-transmitted helminths are generally monitored and evaluated by aggregating drug coverage and infection levels at a district level. However, heterogeneity in drug coverage at finer spatial scales means indicators may remain above thresholds for elimination as a public health problem or of transmission in some areas. This paper aims to highlight the misleading information that aggregating data at larger spatial scales can have for programme decision making.

Methods

Drug coverage data from the Geshiyaro project were compared at two spatial scales with reference to the World Health Organisation's targets. District (woreda) and village (kebele) level were compared. The association between infection levels and drug coverage was analysed by fitting a weighted least-squares function to the mean intensity of infection (eggs per gram of faeces) against drug coverage.

Results

The data show clearly that when the evaluation of coverage is aggregated to the district level, information on heterogeneity at a finer spatial scale is lost. Infection intensity decreases significantly (p = 0.0023) with increasing drug coverage.

Funding: This work was supported by the Children's Investment fund Foundation, UK through a grant to the London Centre for Neglected Tropical Diseases (LCNTDR) at Imperial College London [grant number R-1805-02741]. No authors received a salary from the funders, and the funders took no part in the study design, data collection and analysis, decision to publish or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

Aggregating data at large spatial scales can result in prematurely ceasing deworming, prompting rapid infection bounce-back. There is a strong need to define context-specific spatial scales for monitoring and evaluating intervention programmes.

Author summary

For soil-transmitted helminths (STH), control is centred around mass drug administration (MDA) of eligible subgroups without prior diagnosis, aiming to achieve 75% drug coverage. MDA programmes are monitored by assessing whether drug coverage and infection prevalence and intensity, aggregated to a spatial scale, typically an administrative unit such as a district, have met programmatic targets. Spatial heterogeneity in drug coverage, prevalence, and intensity is commonplace and well documented. However, little attention has been paid to the spatial scales of this heterogeneity and the implications it has on providing the basis to inform the cessation or continuation of prophylactic mass treatment. We present a clear analysis showing that fine scale spatial heterogeneity in drug coverage is commonplace even in well implemented MDA programmes. We show that aggregating data at the district level, which is routine for deworming programmes, can give misleading information on which to base future treatment decisions, specifically the premature reduction in frequency, or cessation of MDA, because fine scale heterogeneity is not captured by larger aggregations. This work highlights the critical need to define context specific spatial scales for monitoring and evaluating interventions implemented in different transmission settings.

Introduction

Soil-transmitted helminths (STH) are a group of intestinal parasites whose mode of transmission is primarily through direct contact, or ingestion, of soil contaminated with infective eggs or larvae [1]. They are geographically distributed in tropical and subtropical regions with focal points of transmission in the most socio-economically deprived populations. Specifically, these most disadvantaged populations are predominantly burdened by the estimated 1.5 billion global infections [2] STHs are targeted for elimination as a public health problem by the World Health Organisation (WHO), defined as achieving less than 2% of moderate or heavy intensity infections. To reach this target, periodic mass drug administration of albendazole or mebendazole to pre-school-aged children (pre-SAC), school-aged children (SAC), women of reproductive age (WRA) and those in what are considered to be high-risk occupations (e.g., agriculturalists for hookworm), is recommended with a coverage target of 75% of the eligible population swallowing the treatment [3,4]

In contexts where MDA is being implemented to eliminate STH as a public health problem, performance indicators (e.g., drug coverage levels) and impact indicators (e.g., prevalence and intensity of infection), are aggregated to a defined spatial scale, which is typically an administrative unit such as a district, as a means of monitoring the intervention [5]. After periodic MDA with good coverage, the prevalence and intensity of infection indicators collected from sentinel sites, aggregated at the district level, are likely to decrease and may indicate that overall prevalence has decreased sufficient enough to reduce the frequency of MDA rounds. However, it is well recognised that spatial heterogeneity in prevalence and intensity of STH infection is

commonplace and as such, hotspots of infection can be important drivers of transmission even when average levels are low [6] Thus, it is highly probable that existing heterogeneity means that in particular areas, at finer spatial scales, the prevalence and intensity are likely to remain above this threshold and the premature altering of MDA strategies may induce a rapid infection bounce-back to pre-intervention levels. Variation in MDA coverage is itself an important driver of observed spatial heterogeneity in prevalence and intensity of STH infection [7] While substantial coverage heterogeneity has been documented [8–10]]. factors driving this variability are not well understood and very little has been discussed about the consequences for control and elimination of using large spatial scales for monitoring and evaluation (M&E). Furthermore, current estimates for the coverage and frequency of MDA are derived from non-spatially structured transmission models which do not consider heterogeneity in coverage between connected geographical areas [6].

Here, we use MDA coverage data collected in year 4 of the Geshiyaro project in Southern Ethiopia and its coverage evaluation surveys (CES) to highlight the importance of needing to define an adequate, context-specific, and possibly dynamic, spatial scale across which to evaluate the progress of interventions aimed at eliminating STH as a public health problem. Factors involved with choosing an appropriate spatial scale are discussed. Detailed descriptions of the overall design and aims of the Geshiyaro project are recorded in previous publications [11]

Methods

Ethics statement

Data were collected as part of the Geshiyaro project, a longitudinal STH elimination feasibility study, the protocol of which has been previously published [11]. The study received ethical approval from the Institutional Review Board (IRB) at the Scientific and Ethical review Office of the Ethiopian Public Health Institute. Formal consent was obtained verbally from all individuals and for children, a parent provided formal verbal consent.

Study site and design

The Geshiyaro project is set in the Wolaita zone of south-western Ethiopia, where there are an estimated two million inhabitants living across 23 districts (woreda). The project has three intervention arms designed to assess which combination(s) of interventions are most effective at reducing STH prevalence. Arm one interventions involve community-wide MDA, improved WaSH infrastructure, and behavioural change communication (BCC), arm two involves community-wide MDA and the existing national 'one WaSH' programme and arm three involves school-based MDA and the existing national 'one WaSH' programme. Detail can be found in the protocol paper previously published [11]. A census was carried out in a sample of five districts (three from arm one and two from arm two) prior to any interventions. All household participants were registered using biometric fingerprint scanning and/or subject ID cards. During subsequent project activities, such as MDA, participants were linked longitudinally by accessing their 11-digit identification number either through biometric fingerprint scanning or from the physical ID card. Individuals not enrolled during the census were newly registered and added to the population census database during these project activities. The subsequent analyses use data collected from the five districts where biometric registration was employed and monitors over 600,000 individuals. Enrolment status did not prevent anyone from accessing deworming medication.

Mass drug administration

Albendazole was intended to be administered annually to all districts of the Wolaita zone, however the delivery method was increased to biannual, to account for delayed MDA rounds during the height of the Covid-19 pandemic. Distribution was also changed from fixed-point to house-to-house to avoid mass gatherings. During each MDA round, all consenting individuals approached by a health extension worker (HEW) were identified by their fingerprint (biometric), subject ID, or name previously recorded during census activities. All individuals were offered treatment regardless of their infection status and individuals who did not consent to having a biometric taken were also offered treatment as usual as part of the national deworming programme. For participants who accepted treatment, the aim was for HEWs to record whether treatment was directly observed. The definition of MDA coverage used herein is the percentage of eligible participants who accepted albendazole.

Spatial data preparation

The delineation of administrative units in Ethiopia is highly variable, particularly in the Southern Nations, Nationalities and Peoples region. Redistricting at all administrative unit levels is commonplace, but the latest changes are not regularly updated and thus reflected in available geo-referenced data. Additionally, there are often discrepancies in the spellings of administrative unit names attributed to the difficulty in translating Amharic and local languages to Latin scripture. Data management alongside current Ministry of Health (MoH) information allowed existing shapefile data (available from https://gadm.org/download_country.html) for the Wolaita zone to be manipulated in such a way that MDA coverage for all current villages (or cluster of villages) could be mapped. In some instances, what was previously a single village cluster (and represented by a single spatial polygon in shapefile data) has now been split into multiple clusters. In such cases, MDA coverage for the villages that make up a single village was aggregated to allow for mapping. In other instances, villages have become town administrations (distinct administrative units in Ethiopia) or been incorporated into new districts, which do not fall under the reach of the Geshiyaro Project. In these instances, data from the coverage evaluation surveys (CES)-designed to provide robust drug coverage estimates and overcome limitations of reported coverage-conducted by Ethiopian Public Health Institute (EPHI) staff following a round of MDA implemented by the Geshiyaro project, were used to obtain complete maps [12]. As a result, CES coverage estimates were generated from a different denominator, yet still representative of the Geshiyaro census denominator. Results clearly indicate the few instances where validated coverage data could not be identified for an administrative unit.

Statistical analyses

All analyses and figures were computed in RStudio (R version 4.1.2). Shapefile data were obtained from the Database of Global Administrative Areas. Participants were grouped into age brackets: Pre-SAC (2 – 4yrs), SAC (5 – 14yrs) and adolescents and adults (>14yrs). For the coverage maps, percentage standard errors were calculated and 95% confidence intervals using the Clopper-Pearson method were calculated for arithmetic mean prevalence. For the association between infection intensity (eggs per gram of faeces [EPG]) and coverage, bootstrapped sampling (r = 1000) with replacement was employed for calculating mean intensity and 95% confidence intervals [13].

Results

Biometrically verified MDA coverage data was available for 92 villages and coverage evaluation survey (CES) survey data was used for the remaining 35 villages that were not under the remit of Geshiyaro MDA. The villages fall within the five censused districts (Fig 1).

There was substantial heterogeneity in Pre-SAC coverage with a range from 40% to 100% and a mean of 82.2% (SE = 2.73) across the entire area at the village level. Overall, a quarter of the villages (33/127) did not meet the 75% coverage targets set out by the WHO for elimination as a public health problem. However, when aggregated to the district level, coverage was >75% for all woredas, with a range of 77.3% to 84.7% and mean of 81.9% (SE = 0.32) (Fig 2).



Fig 1. Map of Ethiopia showing the location of the Geshiyaro study region in southwestern Ethiopia. Coloured polygons in the inset map highlight the five districts that were censused and biometrically monitored. Shapefile data from GADM (https://gadm.org/download_country.html).

https://doi.org/10.1371/journal.pntd.0011947.g001

In SAC, heterogeneity in coverage also existed at the village level, ranging between 45·4% and 100% with a mean of 87·0% (SE = 1·47). In SAC, just over 5% (8/127) of villages did not meet the 75% coverage targets and coverage was >75% in all districts when aggregated at the district level, with a range of 83·6% to 89·6% and a mean of 86·5% (SE = 0·18) (Fig 3).

In adolescents and adults (>14 years), heterogeneity at the village level existed within a narrower range of 58.8% to 85.1% and with a mean of 83.9% (SE = 1.37). Approximately a tenth of villages (12/127) did not meet the 75% coverage targets but surpassed the 75% target when



Fig 2. Maps of the Geshiyaro study region within the Wolaita zone. (a) village level MDA coverage in pre-School Aged Children (Pre-SAC) populations; (b) standard error of MDA coverage; (c) villages that were above or below WHO's target threshold for MDA coverage (75%); (d) the aggregated MDA coverage at the district level. Abbreviations: Pre-SAC-pre-school-aged children. Shapefile data from GADM (https://gadm.org/download_country.html).

https://doi.org/10.1371/journal.pntd.0011947.g002

aggregated at the district level, with a range of $82 \cdot 2\%$ to $87 \cdot 7\%$ and a mean of $84 \cdot 7\%$ (SE = 0.44) (Fig 4).

On average, CES coverage (90.5%; 95% CI: 76.5%– 96.7% tended to be higher than the Geshiyaro's biometrically verified coverage (82.1%; 95% CI: 79.8%– 84.4%), however, the overlapping 95% confidence intervals suggest there was no statistically significant difference between the means. Similarly, comparing the district and age stratified coverage estimates for the two data sources showed overlapping 95% confidence intervals (S1 Fig). As such, this difference was not considered important for further investigation.



Fig 3. Maps of the Geshiyaro study region within the Wolaita zone. (a) village level MDA coverage in School Aged Children (SAC) populations; (b) standard error of MDA coverage; (c) villages that were above or below WHO's target threshold (75%) for MDA coverage; (d) the aggregated MDA coverage at the district level. Abbreviations: SAC-school-aged children. Shapefile data from GADM (https://gadm.org/download_country.html).

https://doi.org/10.1371/journal.pntd.0011947.g003





https://doi.org/10.1371/journal.pntd.0011947.g004

Discussion

Assessing the success of deworming programmes is dependent on an ability to accurately measure key indicators, such as MDA coverage and infection prevalence and intensity. The WHO has set out an MDA coverage target of 75% in pre-SAC, SAC and WRA with the aim of eliminating STH as a public health problem. However, no guidance is given for fine spatial scale reporting of these metrics.

The Geshiyaro project data on MDA coverage shows clearly that when the evaluation of coverage is aggregated up to the district level, information on heterogeneity at a finer spatial scale (i.e., village level) is lost, leading to poorly informed M&E and downstream programmatic efficacy. Panel D in Figs 2-4 illustrate that while coverage exceeds the 75% WHO target in all districts for all age groups, there are obvious pockets of low coverage (40-60%) in the finer spatial scale of villages (Panel A). After four years of MDA, A. lumbricoides infection intensities were all classed as low intensity according to WHO guidance (1-4,999 EPG) [4]. While significant reductions in infection intensities may be made with increasing coverage at the district level, some relatively higher infection intensities likely remain, seeded by villages within a district with lower coverage and, concomitantly, higher infection prevalence and intensity. From a public health policy perspective, the implication of this is that failing to monitor intervention progress at finer spatial scales means national control programs could prematurely reduce or stop MDA due to the false conclusion that coverage targets aggregated at less granular spatial scales have been uniformly met across small spatial scale units such as a village(s). For example, the Expanded Special Project for Elimination of Neglected Tropical Diseases (ESPEN), which was established by the WHO Regional Office for Africa to accelerate elimination of the five most prevalent NTDs, summarises data at the district level which may inadvertently suggest more progress has been made towards these goals than is strictly the case if fine scale pockets of high infection persist. Small pockets of medium or high infection prevalence and intensity in a landscape of low average infection levels can potentially sustain transmission across much larger spatial scales. The constraints of aggregating data are suitably shown by Truscott et al [14,15] in data analysis for the DW3 project on STH control in three countries [16] who found that while the mean prevalence of a cluster of communities may be below a given threshold, some individual communities may be above it and this decreased the likelihood of interrupting transmission. In fact, based on parameter estimates from the DW3 study, the stochastic individual based simulations showed that a collection of communities in a cluster failing to reach the elimination threshold, and with prevalence above the cluster mean, will likely bounce back to endemicity, resulting in the whole cluster failing to interrupt transmission as predicted [14]

Ideally, M&E would be carried out at the smallest administrative unit, however, this is resource intensive so to a large extent, funding availability dictates how fine a scale M&E can be carried out at. Nonetheless, a future strategy that involves identifying hotspots of high prevalence driven by high rates of reinfection [17]. and targeting MDA to these areas means performance and impact indicators can be monitored at the finest scale possible, removing the need for a blanket, resource intensive M&E strategy. Hotspots of high prevalence are important for sustaining transmission [15,18,19] and thus if they could be prioritised for MDA, the opportunities for elimination, whether as a public health problem or transmission interruption, are increased. This practice has already begun to be implemented for schistosomiasis [20,21]. Recently a framework of diagnostic use-cases and target product profiles (TPP) was developed to categorise transitions between control programme phases [22,23]. A similar framework should be developed that defines adequate spatial scales for M&E as programmes move from morbidity control to elimination as transmission interruption, using hotspots that need treatment targeting to define the *use-case* for programmes in low transmission settings.

As countries shift their focus to the possibility of eliminating transmission, a critical question that remains unanswered is what is an appropriate spatial unit to define STH transmission within and between adjacent units? For a clearly defined spatial unit with a known pristine transmission intensity level prior to widescale MDA, a formal coverage target should exist for the elimination or cessation of parasite transmission. At present none exists. Based on mathematical models of parasite transmission and MDA impact with parameters derived from data [15] the Geshiyaro project, which aims to assess the feasibility of transmission interruption, has set a drug coverage target of 90% across all age groups to achieve this goal. Nevertheless, Anderson and others [14,15,19] note the ambiguities surrounding these proposed threshold targets. For example, transmission interruption depends on the initial baseline prevalence and average intensity of infection (both determined by the magnitude of the basic reproductive number R_0). Furthermore, regardless of the initial transmission intensity, achieving transmission interruption with prediction models of parasite transmission and MDA impact occurs only in the absence of movement of infected individuals from neighbouring villages or districts who can shed infective stages into the environment [14,19]. In reality this assumption does not hold true. In a given setting, the spatial unit over which transmission occurs depends on many factors including urbanisation, transportation routes, location of schools and health care facilities, and human mobility. The likelihood that parasites will be reintroduced following successful transmission interruption in a defined location will depend on the linkages of the area with other locations which may or may not have reached the conditions required to stop transmission. The influence of these factors will vary greatly in different geographical and cultural settings. Our understanding of these factors and variation of them in different contexts is very limited at present. This may help explain why WHO policy on defining what spatial scales MDA coverage levels should be reported is so vague at present. What can be done to improve knowledge in this area?

Whilst non-spatially structured transmission models can propose probabilities of elimination or bounce-back under various scenarios [15,24] the reality of identifying transmission breakpoints in different settings must include the coupling of spatial movement patterns with parasite genetic data and molecular epidemiological analyses. Identifying variable genome regions creates the opportunity to ascertain 'who infects whom' over different spatial scales, for example, within households, within and between households in a village, and between villages Add citation. Sampling the genetic material of expelled adult STH worms post-MDA via worm expulsion studies and associated phylogenetic analyses, offers the ability to describe the STH genomic landscape in a geographical region [25,26]

Recent stochastic mathematical modelling studies of STH transmission and control within a spatially heterogeneous environment in Kenya [27] show that the correct spatial scale at which to evaluate intervention progress does indeed depend on the movement patterns of people as linked to the spatial dispersal of infective stages in defined social settings, as does the likelihood and degree to which transmission will persist. Our understanding of how STH transmission, as well other NTDs, is affected by spatial scale is very poor at present but molecular epidemiological studies would act to remedy this situation.

On average, the villages achieving the highest coverage tended to be where CES data was used instead of Geshiyaro data (Figs 2–4). Without an in-depth analysis, it is unclear whether the CES methodology [4] generated an overestimation of coverage. The general concern for survey-based data collection as suggested by WHO for the CES surveys, is treatment recall bias from participants, and health workers exaggerating coverage which can both lead to overestimation [28,29].

One limitation of this study arises from the two different data sources used to generate coverage maps. On the one hand, MDA coverage from Geshiyaro data is biometrically verified and based on directly observed treatment, whilst CES coverage is based on participants selfreporting drug swallowing, which is known to be subject to substantial bias. Moreover, the denominators used for the coverage calculations of each data source differ; Geshiyaro denominators represent the entire eligible population for an administrative unit, whilst CES denominators represent the total number of participants surveyed, i.e., a sub-sample. Nonetheless, the population targeted for CES are all community members living in the five biometric woredas who are eligible for treatment. Thus, the CES denominator can be considered a representative sample of the Geshiyaro denominator and therefore coverage data similar enough to display concomitantly with Geshiyaro coverage.

Conclusion

The main conclusion of this study is that aggregated data on MDA coverage and associated infection levels can give misleading information on which to base decisions on whether or not to stop prophylactic mass treatment to control STH infections. Finer spatial scale data reveals great heterogeneity, even when the overall mean coverage is high. The appropriate scale on which to report data and judge success, or failure, will vary by infection type and social setting. Both influence the scale on which transmission takes place and with what frequency. For many non-vector borne NTDS-the most important scale is likely to be within household and within village. However, the importance of between village and within district transmission is very poorly understood for most parasitic infections. Molecular epidemiological methods could help to reveal more about 'who infects whom' but to date these have not been employed in epidemiological research on the human helminth parasites.

Supporting information

S1 Fig. Comparison of MDA coverage estimates from the two data sources stratified by district and age group. (PDF)

Acknowledgments

We extend our thanks to all the health extension workers and zonal health authorities for their efforts in conducting field work and to the communities for their time and participation.

Author Contributions

Conceptualization: Santiago Rayment Gomez, Roy Anderson.

Data curation: Santiago Rayment Gomez, Rosie Maddren, Melkie Chernet.

Formal analysis: Santiago Rayment Gomez, Benjamin Collyer.

Investigation: Ewnetu Firdawek Liyew, Ufaysa Anjulo, Adugna Tamiru.

Methodology: Santiago Rayment Gomez.

Supervision: Roy Anderson.

Validation: Benjamin Collyer.

Visualization: Santiago Rayment Gomez.

Writing – original draft: Santiago Rayment Gomez, Roy Anderson.

Writing – review & editing: Santiago Rayment Gomez, Rosie Maddren, Ewnetu Firdawek Liyew, Melkie Chernet, Ufaysa Anjulo, Adugna Tamiru, Getachew Tollera, Geremew Tasew, Birhan Mengistu, Benjamin Collyer, Kathryn Forbes, Roy Anderson.

References

 Brooker SJ, Nikolay B, Balabanova D, Pullan RL. Global feasibility assessment of interrupting the transmission of soil-transmitted helminths: a statistical modelling study. Lancet Infect Dis. 2015; 15(8):941– 50. https://doi.org/10.1016/S1473-3099(15)70042-3 PMID: 25886799

- Jourdan PM, Lamberton PHL, Fenwick A, Addiss DG. Soil-transmitted helminth infections. The Lancet. 2018 Jan 20; 391(10117):252–65. <u>https://doi.org/10.1016/S0140-6736(17)31930-X</u> PMID: 28882382
- WHO. Helminth control in school-age children: a guide for managers of control programmes [Internet]. 2011. https://www.who.int/publications/i/item/9789241548267
- WHO. Geneva: World Health Organization; 2017. 2017. Guideline: Preventative chemotherapy to control soil-transmitted helminth infections in at-risk population groups. <u>https://apo.who.int/publications/i/</u> item/9789241550116
- 5. WHO. ELIMINATING SOIL-TRANSMITTED HELMINTHIASES AS A PUBLIC HEALTH PROBLEM IN CHILDREN: PROGRESS REPORT 2001–2010. 2012.
- Vegvari C, Truscott JE, Kura K, Anderson RM. Human population movement can impede the elimination of soil-transmitted helminth transmission in regions with heterogeneity in mass drug administration coverage and transmission potential between villages: a metapopulation analysis. Parasit Vectors. 2019 Sep; 12(1):438. https://doi.org/10.1186/s13071-019-3612-7 PMID: 31522681
- Truscott JE, Ower AK, Werkman M, Halliday K, Oswald WE, Gichuki PM, et al. Heterogeneity in transmission parameters of hookworm infection within the baseline data from the TUMIKIA study in Kenya. Parasit Vectors. 2019; 12(1):1–13.
- Parker M, Allen T. Does mass drug administration for the integrated treatment of neglected tropical diseases really work? Assessing evidence for the control of schistosomiasis and soil-transmitted helminths in Uganda. Health Res Policy Syst. 2011; 9(1):3. https://doi.org/10.1186/1478-4505-9-3 PMID: 21211001
- Okoyo C, Campbell SJ, Minnery M, Owaga C, Onyango N, Medley G, et al. Prevalence and Correlation Analysis of Soil-Transmitted Helminths Infections and Treatment Coverage for Preschool and School Aged Children in Kenya: Secondary Analysis of the National School Based Deworming Program Data. Front Public Health [Internet]. 2021; 9. Available from: https://www.frontiersin.org/articles/10.3389/ fpubh.2021.645522 PMID: 34336756
- Bah YM, Bah MS, Paye J, Conteh A, Saffa S, Tia A, et al. Soil-transmitted helminth infection in school age children in Sierra Leone after a decade of preventive chemotherapy interventions. Infect Dis Poverty [Internet]. 2019; 8(1):41. Available from: <u>https://doi.org/10.1186/s40249-019-0553-5</u> PMID: 31262367
- Mekete K, Ower AK, Dunn JC, Sime H, Tadesse G, Abate E, et al. The Geshiyaro Project: a study protocol for developing a scalable model of interventions for moving towards the interruption of the transmission of soil-transmitted helminths and schistosome infections in the Wolaita zone of Ethiopia. Parasit Vectors. 2019; 12.
- Liyew EF, Chernet M, Belay H, Maddren R, Landeryou T, Kalahasti S, et al. Coverage evaluation surveys following soil-transmitted helminthiasis and schistosomiasis mass drug administration in Wolaita Zone of Ethiopia-The Geshiyaro project. PLoS One. 2021; 16(12):e0260722. <u>https://doi.org/10.1371/journal.pone.0260722</u> PMID: 34932602
- Werkman M, Wright JE, Truscott JE, Oswald WE, Halliday KE, Papaiakovou M, et al. The impact of community-wide, mass drug administration on aggregation of soil-transmitted helminth infection in human host populations. Parasit Vectors [Internet]. 2020; 13(1):290. Available from: https://doi.org/10. 1186/s13071-020-04149-4 PMID: 32513254
- Truscott JE, Werkman M, Wright JE, Farrell SH, Sarkar R, Ásbjörnsdóttir K, et al. Identifying optimal threshold statistics for elimination of hookworm using a stochastic simulation model. Parasit Vectors [Internet]. 2017; 10(1):321. Available from: <u>https://doi.org/10.1186/s13071-017-2256-8</u> PMID: 28666452
- Anderson R, Truscott J, Hollingsworth TD. The coverage and frequency of mass drug administration required to eliminate persistent transmission of soil-transmitted helminths. Philos Trans R Soc Lond B Biol Sci [Internet]. 2014; 369(1645):20130435. Available from: http://europepmc.org/abstract/MED/ 24821921 https://doi.org/10.1098/rstb.2013.0435
- Ásbjörnsdóttir KH, Ajjampur SSR, Anderson RM, Bailey R, Gardiner I, Halliday KE, et al. Assessing the feasibility of interrupting the transmission of soil-transmitted helminths through mass drug administration: The DeWorm3 cluster randomized trial protocol. PLoS Negl Trop Dis [Internet]. 2018 Jan 18; 12 (1):e0006166. Available from: https://doi.org/10.1371/journal.pntd.0006166 PMID: 29346377
- 17. Rayment Gomez S, Maddren R, Liyew EF, Chernet M, Anjulo U, Tamiru A, et al. Predisposition to soil-transmitted helminth reinfection after four rounds of mass drug administration: results from a longitudinal cohort in the Geshiyaro project, a transmission elimination feasibility study in the Wolaita zone of southern Ethiopia. Trans R Soc Trop Med Hyg. 2023 Mar 20;trad007.
- 18. Anderson RM, Turner HC, Truscott JE, Hollingsworth TD, Brooker SJ. Should the Goal for the Treatment of Soil Transmitted Helminth (STH) Infections Be Changed from Morbidity Control in Children to

Community-Wide Transmission Elimination? PLoS Negl Trop Dis. 2015 Aug 20; 9(8):e0003897. https://doi.org/10.1371/journal.pntd.0003897 PMID: 26291538

- Truscott JE, Hollingsworth TD, Brooker SJ, Anderson RM. Can chemotherapy alone eliminate the transmission of soil transmitted helminths? Parasit Vectors. 2014 Jun; 7:266. <u>https://doi.org/10.1186/</u> 1756-3305-7-266 PMID: 24916278
- Amoah AS, Hoekstra PT, Casacuberta-Partal M, Coffeng LE, Corstjens PLAM, Greco B, et al. Sensitive diagnostic tools and targeted drug administration strategies are needed to eliminate schistosomiasis. Lancet Infect Dis [Internet]. 2020; 20(7):e165–72. Available from: https://www.sciencedirect.com/ science/article/pii/S1473309920302541 https://doi.org/10.1016/S1473-3099(20)30254-1 PMID: 32595046
- Tchuem Tchuenté LA, Stothard JR, Rollinson D, Reinhard-Rupp J. Precision mapping: An innovative tool and way forward to shrink the map, better target interventions, and accelerate toward the elimination of schistosomiasis. PLoS Negl Trop Dis [Internet]. 2018 Aug 2; 12(8):e0006563-. Available from: https://doi.org/10.1371/journal.pntd.0006563 PMID: 30071014
- 22. Solomon AW, Engels D, Bailey RL, Blake IM, Brooker S, Chen JX, et al. A Diagnostics Platform for the Integrated Mapping, Monitoring, and Surveillance of Neglected Tropical Diseases: Rationale and Target Product Profiles. PLoS Negl Trop Dis [Internet]. 2012 Jul 31; 6(7):e1746-. Available from: <u>https://doi.org/10.1371/journal.pntd.0001746 PMID: 22860146</u>
- Lim MD, Brooker SJ, Belizario VY Jr., Gay-Andrieu F, Gilleard J, Levecke B, et al. Diagnostic tools for soil-transmitted helminths control and elimination programs: A pathway for diagnostic product development. PLoS Negl Trop Dis [Internet]. 2018 Mar 1; 12(3):e0006213-. Available from: <u>https://doi.org/10. 1371/journal.pntd.0006213</u> PMID: 29494581
- Truscott JE, Turner HC, Farrell SH, Anderson RM. Chapter Three—Soil-Transmitted Helminths: Mathematical Models of Transmission, the Impact of Mass Drug Administration and Transmission Elimination Criteria. In: Basáñez MG, Anderson RM, editors. Advances in Parasitology [Internet]. Academic Press; 2016. p. 133–98. https://www.sciencedirect.com/science/article/pii/S0065308X1630077X
- 25. Easton A V, Oliveira RG, O'Connell EM, Kepha S, Mwandawiro CS, Njenga SM, et al. Multi-parallel qPCR provides increased sensitivity and diagnostic breadth for gastrointestinal parasites of humans: field-based inferences on the impact of mass deworming. Parasit Vectors [Internet]. 2016; 9(1):38. Available from: https://doi.org/10.1186/s13071-016-1314-y PMID: 26813411
- Levecke B, Easton A V, Cools P, Albonico M, Ame S, Gilleard JS, et al. The optimal timing of post-treatment sampling for the assessment of anthelminthic drug efficacy against Ascaris infections in humans. Int J Parasitol Drugs Drug Resist [Internet]. 2018; 8(1):67–9. Available from: https://www.sciencedirect. com/science/article/pii/S2211320717301367 https://doi.org/10.1016/j.ijpddr.2017.12.004 PMID: 29414108
- 27. Collyer BS, Truscott JE, Mwandawiro CS, Njenga SM, Anderson RM. How important is the spatial movement of people in attempts to eliminate the transmission of human helminth infections by mass drug administration? Philos Trans R Soc Lond B Biol Sci. 2023 Oct 9; 378(1887):20220273. Available from: https://doi.org/10.1098/rstb.2022.0273 PMID: 37598700
- Sheahan W, Anderson R, Aruldas K, Avokpaho E, Galagan S, Goodman J, et al. Overestimation of school-based deworming coverage resulting from school-based reporting. PLoS Negl Trop Dis. 2023 Apr 10; 17(4):e0010401. https://doi.org/10.1371/journal.pntd.0010401 PMID: 37036890
- 29. Onkanga IO, Mwinzi PNM, Muchiri G, Andiego K, Omedo M, Karanja DMS, et al. Impact of two rounds of praziquantel mass drug administration on Schistosoma mansoni infection prevalence and intensity: a comparison between community wide treatment and school based treatment in western Kenya. Int J Parasitol. 2016 Jun; 46(7):439–45. https://doi.org/10.1016/j.ijpara.2016.01.006 PMID: 26940547